Author's response to reviews

Title: Mortality attributable to tobacco: Review of different methods

Authors:

Nabil Tachfouti (tachfouinabil@yahoo.fr)
Chantale Raherison (chantal.raherison@chu-bordeaux.fr)
Majdouline Oubtel (majdoctel7@yahoo.fr)
Chakib Nejjari (cnejjari2000@yahoo.fr)

Version: 6 Date: 23 February 2014

Author's response to reviews:

To
Dr. Herman Van Oyen
Editor-in-Chief
Archive of public health

Ref: your mail Re: 1566326679108104. Mortality attributable to tobacco: Review of different methods

Sir,

Please find enclosed the revised manuscript now titled “Mortality attributable to tobacco: Review of different methods”

We highly appreciate these comments for their usefulness in helping us make the content of the article more precise and clearer. Following are the detailed responses to these comments.

Editorial comments:

1. abstract:
the general rate of Smoking attributable mortality (SAM) this should be changed to the general rate of Smoking Attributable Mortality (SAM)
R. Done

2. Table 1 - Methods used in calculating the SAM by location and date of study change to Table 1. Methods used in calculating the Smoking Attributable Mortality (SAM) by location and date of study
R. Done

3. Table 2 - ICD 9-CM and ICD 10 codes for smoking related diseases change to Table 2. ICD 9-CM and ICD 10 codes for smoking related diseases
R. Done

4. Table 3: Summary of results reported in a review change to Table 3. Summary
of Smoking Attributable Mortality (SAM) as reported in the articles included in the review.
R. Done